Author’s response to reviews

Title: Menstrual Health and Hygiene among Indigenous Australian girls and women: Barriers and opportunities

Authors:

Emily Krusz (e.krusz@uq.edu.au)
Nina Hall (n.hall2@uq.edu.au)
Dani Barrington (d.j.barrington@leeds.ac.uk)
Sandra Creamer (s.creamer@uq.edu.au)
Wendy Anders (seniorprojects@natsiwa.org.au)
Minnie King (womenoncountry@bigpond.com)
Helen Martin (womenoncountry1@bigpond.com)
Julie Hennegan (jhenneg1@jhu.edu)

Version: 4 Date: 01 Oct 2019

Author’s response to reviews:

Dear editors,

Thank you for your helpful guidance to finalise this manuscript.

In response to your comments, we have:

1. Consent to participate
   Clarified the written consent provided by participants.

2. Role of Funders
   Clarified the non-involvement if te funders in data enaggement.

3. Clean manuscript
   Uploaded a clean copy.
Please let us know if further changes are required.

Thank you,

Nina Hall